Author's response to reviews

Title: Acute cholecystitis with massive upper gastrointestinal bleed: A case report and review of the literature

Authors:

Sundeep S Saluja (sundepsaluja@yahoo.co.in)
Sukanta Ray (drsukantaray@yahoo.co.in)
Manpreet S Gulati (m_gulati@hotmail.com)
Sujoy Pal (sujoypal@hotmail.com)
Peush Sahni (peush_sahni@hotmail.com)
Tushar K Chattopadhyay (tkc46@hotmail.com)

Version: 6 Date: 23 March 2007

Author's response to reviews:

23 rd March 2007
Annabel Phillips PhD,
Assistant Editor, BMC-series Journals
BioMed Central, Middlesex House
34-42 Cleveland Street, London W1T 4LB

Subject: Resubmission of manuscript no MS 1634762455118696 for publication in BMC Gastroenterology.
Dear Sir,
I am hereby resubmitting the manuscript entitled 'Acute cholecystitis with massive upper gastrointestinal bleed: a case report and review of the literature' to after revising according to the formatting changes advised.
1 All authors have contributed equally has been removed from title page and the contribution of each author is mentioned in the authors' contribution section.
2 Reference numbers have arranged in the proper order within square brackets
3 Author's contribution changed to authors’ contribution
4 Figure Legends changed to Figure legends
5 Figures cropped accordingly
6 There is no funding agency involved in the work

Yours sincerely,
Sundeep Singh Saluja
Senior Research Associate and Corresponding author,
Room No 1005, PC block, 1st floor, Department of Gastrointestinal Surgery, All India Institute of Medical Sciences, Ansari Nagar, New Delhi-110029, India.
Email: sundepsaluja@yahoo.co.in; Fax: 91-11-26588663